Development of the Nairobi Stock Exchange: A Historical Perspective

Rose. W. Ngugi

Macroeconomics Division Kenya Institute for Public Policy Research and Analysis

KIPPRA Discussion Paper No. 27 March 2003

KIPPRA IN BRIEF

The Kenya Institute for Public Policy Research and Analysis (KIPPRA) is an autonomous institute whose primary mission is to conduct public policy research, leading to policy advice. KIPPRA's mission is to produce consistently high-quality analysis of key issues of public policy and to contribute to the achievement of national long-term development objectives by positively influencing the decision-making process. These goals are met through effective dissemination of recommendations resulting from analysis and by training policy analysts in the public sector. KIPPRA therefore produces a body of well-researched and documented information on public policy, and in the process assists in formulating long-term strategic perspectives. KIPPRA serves as a centralized source from which the government and the private sector may obtain information and advice on public policy issues.

Published 2003

© Kenya Institute for Public Policy Research and Analysis Bishops Garden Towers, Bishops Road PO Box 56445, Nairobi, Kenya tel: +254 20 2719933/4; fax: +254 20 2719951

email: admin@kippra.or.ke
website: http://www.kippra.org

ISBN 9966 949 48 8

The Discussion Paper Series disseminates results and reflections from ongoing research activities of the institute's programmes. The papers are internally refereed and are disseminated to inform and invoke debate on policy issues. Opinions expressed in the papers are entirely those of the author or authors and do not necessarily reflect the views of the Institute.

KIPPRA acknowledges generous support by the European Union (EU), the African Capacity Building Foundation (ACBF), the United States Agency for International Development (USAID), the Department for International Development of the United Kingdom (DfID) and the Government of Kenya (GoK).

ABSTRACT

This paper explores the evolutionary process of the stock market in Kenya and identifies the institutional and policy changes that have shaped the development pattern of the Nairobi Stock Exchange (NSE). A historical perspective approach is used to trace developments in the regulatory system, trading system, market membership, composition of market investors and taxation policy. Three phases in the development process of the stock market are identified and distinguished by their unique institutional and policy environment characteristics. These phases are the initiation stage, the formalization stage and the revitalization/ restructuring stage. The initial stage is mainly characterized by dominations of foreign investors in share trading and spontaneous growth. The formalization stage saw the adoption of a self-regulatory framework with efforts to increase the participation of local citizens in share trading especially in the postindependence period. This is also the period when the Government adopted a controlled policy regime and implemented tax policies that penalized share returns more than returns from other financial assets. Further, the Nairobi Stock Exchange which had served as a regional market among the East African states lost its market share with the break-up of the East African Community. Partially because of these developments, the stock market was characterized by a state of dormancy during this phase. This gave way to the revitalization stage where efforts were made to strengthen the institutional infrastructure and also to enhance the policy environment in order to facilitate growth of the stock market.

Abbreviations

AIM Alternative Investment Market AKS Association of Kenya Stockbrokers

ATS Automated Trading System
CDS Central Depositing System
CIC Capital Issue Committee
CMA Capital Markets Authority

CMDC Capital Market Development Committee

DvP Delivery versus Payment (system)

EAC East African Community

EADB East African Development Bank
EALA East African Legislative Assembly

EASRA East African Member State Securities Regulatory

Authorities

FISM Fixed Income Securities Market IFC International Finance Corporation

IPO Initial Public Offer
LSE London Stock Exchange
MIM Main Investment Market
NSE Nairobi Stock Exchange
NSSF National Social Security Fund
RBA Retirement Benefits Authority
SEC Security Economic Commission

USAID United States Agency for International Development

TABLE OF CONTENTS

1.	Intro	duction1				
2.	Pre-reform period: 1920-1989 4					
	2.1	Stage I (1920s-1953): Inception 4				
	2.2	Stage II (1954-1963): formalization of share trading 5				
	2.3	Stage III (1963-1970): Political environment and Kenyanisation policy				
	2.4	Stage IV (1971-1989): Capital Issue Committee and taxation policy				
3.	Revit	alization process				
	3.1	Statutory regulatory framework				
	3.2	Trading system				
	3.3	NSE membership				
	3.4	Foreign investors				
	3.5	New listings				
	3.6	Regional market				
	3.7	Securities				
	3.8	Taxation policy				
4.	Conclusion 40					
	References					
	Appendix					

1. Introduction

The Nairobi Stock Exchange (NSE) has a long history that can be traced to the 1920's when it started trading in shares while Kenya was still a British colony (IFC/CBK, 1984). While share trading was initially conducted in an informal market, there was a growing desire to have a formal market that would facilitate access to long-term capital by private enterprises and also allow commencement of floating of local registered Government loans. The NSE was constituted in 1954 as a voluntary association of stockbrokers registered under the Societies Act (NSE, 1997a). The newly established stock exchange was charged with the responsibility of developing the stock market and regulating trading activities. Despite its history, however, the stock market is yet to make significant contribution in the development process. The question of interest to research is what defines the development path of the stock market? Does it mimic development paths of other developed or emerging markets?

The development path of stock markets in both the emerging and developed world indicates an evolutionary process where changes in institutional infrastructure and the policy environment are witnessed as efforts are made to facilitate the growth of the stock market. The evolutionary process indicates graduation from non-formal markets to formal organizations without a regulatory body and then establishment of a statutory body in the reform/restructuring process. The establishment of a statutory body is aimed at enhancing the confidence of investors. While statutory regulatory bodies in most developed markets are set up to resolve the conflict of interest in the self-regulation framework, most of the emerging markets are establishing such bodies as part of the revitalization reform process.

It is argued that protection of investors encourages participation in share trading and enhances the development of financial markets (La Porta

et al, 1998). It is also indicated that countries that protect shareholders have more valuable stock markets, large number of listed securities per capita and a high rate of Initial Public Offers (IPOs). Further, it is observed that protection of investors and quality of law enforcement as proxy for governance have predictors power on the extent of market declines during crises and even better than macroeconomic variables (La Porta et al, 2000). Ensuring investors' confidence enhances investors' participation in the market activities and encourages saving and channeling of savings into productive real investment, therefore fostering capital accumulation and efficiency in investment and real sector development. It is however debatable whether protection of investors promotes market efficiency.

In addition, it is argued that information disclosure enhances market efficiency by providing informed traders with costless information that ensures market efficiency (Georgakopoulos, 1996). For example, it is observed that securities may deviate from their fundamental values because of irrationalities in the market, including uninformed trading or noise trading. This type of trading interferes with the optimality of market allocation of capital among competing firms. Therefore, it is important that disclosure rules are tightened, especially for the benefit of transient institutional investors who require more accurate prices.

The evolutionary process is also characterized by a shift in trading system from a periodic auction system to a continuous trading system. Trading system defines the price discovery process or the transformation of latent demand of investors into realized transactions (Madhavan, 1992). The evolutionary process of trading system also indicates a shift from manual to electronic and centralized settlement clearing.¹ It is

⁻

¹ Garman (1976) observes an evolutionary pattern in adoption of trading system for the US stock market in response to growth in trading volume; this saw a shift from periodic to continuous trading system. Amihud *et al* (1997) note the tendency for emerging markets to shift from periodic to continuous trading in the revitalization process.

argued that a trading system that enhances efficiency in the price discovery process, provides liquidity at low costs, and has no excess volatility is more desirable for the development of the stock market (Amihud, *et al* 1990, and Bessembinder and Kaufman, 1997). High liquidity, it is observed, enhances long-term investment by reducing the required rate of return and by lowering the cost of capital to the issuers of securities. An efficient price discovery process enhances the role of the market in aggregating and conveying information through price signals, therefore making prices more informative.

This paper traces the development path of the Kenya stock market focusing on the evolutionary process of the Nairobi Stock Exchange (NSE). The main motivation of this paper is the indicated relationship between policy environment and adequacy of the institutional infrastructure and the growth path of stock markets. A historical perspective approach is used in tracing developments in the institutional infrastructure and policy framework. By understanding the development path of the stock market, this study serves as a pointer to the factors that explain the observed performance of the stock market.

The remainder of this paper is structured into four sections. Section 2 covers the period before the formalization of the regulatory system, while section 3 covers the reform process. Section 4 provides a discussion on the response of the market to the various changes while section five provides conclusions that can be drawn from this analysis of the development of the Nairobi Stock Exchange.

2. Pre-reform Period: 1920s-1989

This period preceded the revitalization reforms and can be divided into various development stages defined by specific institutional characteristics. Stage one is the initiation period before formalization of share trading (1920s-1953) while stage two (1954-1963) is the period when the market was formalized but before the political independence. The post-independence period which constitutes stage three (1964-1970) was the period before the Government made the first attempt to regulate the operations of the market. During stage four, the Government established the Capital Issue Committee (CIC) as a first attempt to oversee the operations of the stock market. These four stages are analysed in order to identify the major institutional and policy differences during the various stages of the development of the NSE.

2.1 Stage I (1920s-1953): Inception

Share trading was initiated in the 1920s when Kenya was a British colony. Stage I, the initiation stage of the NSE, covers the period before the establishment of a formal stock market. This period is mainly characterized by informal share trading with no formal rules or regulations to govern trading activities (IFC/CBK, 1984). Trading in shares was based on a gentleman's agreement where standard commissions were charged and clients were obliged to honor their contractual commitment of making good delivery and settling of relevant costs. There was no physical trading floor or specialized stockbrokers. Share trading was a part time job for accountants, auctioneers, estate agents and lawyers who met to exchange prices over a cup of coffee (Parkinson, 1984 and Munga, 1974). The first stock brokerage firm was established in 1951 by Francis Drummond. Foreign investors dominated share trading mainly because they had the knowhow of operating organized capital markets and also because their high

income sufficiently permitted them to accumulate savings and investment in securities (Aworolo, 1971). Participation of local citizens was very limited, accounting for about 5%, mainly because of their low-income and statutory restrictions during the pre-independence period.

2.2 Stage II (1954-1963): Formalization of Share Trading

The main feature of this period is the establishment of the NSE which marked the formalization of share trading. The desire to establish a formal market was initiated by stockbrokers who desired to have a stock exchange that facilitated access by private enterprises to long-term capital. In addition, the Minister for Finance desired to have a formal market that facilitated floating of locally-registered Government loans, which would be unattractive without a stock exchange. This saw the constitution of the Nairobi Stock Exchange in 1954 as a voluntary association of stockbrokers registered under the Societies Act (NSE, 1997a). To facilitate the registration, stockbrokers obtained clearance from the London Stock Exchange (LSE), which recognized the NSE as an overseas stock exchange, effectively enabling the NSE to gain value and credibility (Munga, 1974). The newly established stock exchange was charged with the responsibility of developing the stock market and regulating trading activities.²

_

² The main objectives of NSE (1954) are: improving the facilities available for purchase and sale of shares; regulating the dealings of members with clients, with non-members engaged in stockbroking; standardising and reviewing charges made by members for services rendered to their clients or modify the method(s) of assessing or calculating such charges; correlating stockbroking activities of members and facilitate exchange of information with clients and to the public; co-operating with associations of stockbrokers and stock exchanges to obtain and make available to members information likely to be of advantage to them and clients; investigating, resolving, and taking steps to enforce its decisions and awards relating to any irregularities or alleged irregularities in the dealings of members with clients, with non-members engaged in stockbroking, and any disputes between members and non-members and any complaints against members by other members or any other parties, provided such disputes or complaints relate to or touch on the stockbroking business or activities of such members, among others.

With the establishment of the NSE in 1954, a self-regulatory framework was adopted whose main responsibility was to develop the stock market. The self-regulatory framework, which borrows heavily from the London Stock Exchange (LSE), is embodied in the Rules and Regulations of NSE 1954. The regulatory framework spelt out guidelines for primary issues, the operations of stockbrokers, and secondary market trading activities with the aim of overseeing the development of the stock market. To enforce the rules and regulations, the Exchange vested in a committee of five the powers to govern and manage the Exchange. Members of the committee had to have at least two years membership in the Stock Exchange. The committee was to ensure smooth trading activities by overseeing the activities of members, investors and the listing companies. The committee for example had powers to forbid dealings in shares of any company without giving a reason for such decision. It also had powers to approve public quotations. The committee heard and adjudicated at a fee all disputes between members and complaints against members. It also adjudicated any transactions or disputes or complaints arising out of any transaction in shares entered into by a member, even if such shares may not have received a quotation. Further, the committee provided information, for example the share prices dealt during the previous week, for publication in circulars, newspapers and other mass media.

Although the formal market opened its doors in 1954, it had no physical trading floor and no specialist. Periodic (call) auction³ trading system,

-

³ Such a system operates like a Walrasian Tatoonement for auctions. Price adjustment begins when the board is called and stockbrokers in possession of securities express their interest by signalling a bid or ask based on the perceived demand and supply and clients' instructions. When a stockbroker has buy and sell instructions he signals a sell order with higher price and a buy order with lower price. When a stockbroker wishes to execute a cross deal he signals a single price for both bid and ask and bargain slips are used as evidence of deals stock between members and also serves as memorandum for the transactions inter-brokers delivery and settlement. Theoretically, call auction market is said to be more efficient than the continuous market with its ability to enhance liquidity and reduce market volatility (Madhavan, 1992).

a variant of order-driven trading system, was adopted where transaction prices were exchanged in coffee-house forum. Business was transacted by telephone and prices determined through negotiation. No dealings were recognized except those for NSE members and fully paid shares. Brokers were free to buy or sell shares of their customers without consulting each other, provided such transactions were effected at the best possible price. Representatives of member firms met everyday for a call-over where only the active securities were called; highest limit order to buy (bid) and lowest order to sell (offer) were recorded in call-over sheet. Bid and ask prices for all securities were recorded every Thursday together with the transaction prices following the weekly call-over but only the highest prices were published. Customers were able to buy or sell shares at any time during the business day at prices established at the daily meeting.

Securities traded in the NSE during the period mainly included Government stocks, loan stocks, preferential and common share. Data available from the yearbooks indicate that loan stocks were unpopular compared to the common and preferential shares. In 1954 for example, two companies issued loan stocks worth Kshs 7.48 million as compared to 28 companies which offered preferential shares worth Kshs 90 million. Issues of Government stocks were made on the London Stock Exchange. The value of stocks increased from Kshs 84.88 million in 1953 to Kshs 95.5 million in 1954 following a listing of 12 public sector stocks and Kshs 295.5 million in 1959. When the NSE started operations, only 5.4% of Government stocks were listed; the rest had listing in the London Stock Exchange (LSE).

The number of stockbrokers increased to six in 1954, with two specialists and others carrying out stockbroking as subsidiary activities (Parkinson, 1984). In the *Rules and Regulations of NSE 1954* all individuals, firms and limited liability companies transacting brokerage business and whose applications had the approval of the NSE Committee are regarded

as members of the Stock Exchange. However, to be a member, stockbrokers had to pay an entrance fee and an annual subscription fee. In addition, stockbrokers had to have at least three years working experience at senior level with a registered member of the NSE or any other recognized stock exchange. They also had to deposit with the NSE Kshs 50,000, which was refundable without interest at the expiry of probation period of at least one year as indicated in Table 1.

Stockbrokers acted as agents, buying and selling securities for their clients. They were also expected to provide information and advice to the clients and help firms (in conjunction with the underwriter) in determining a fair issue price. The *Rules and Regulations of NSE 1954* allowed stockbrokers to perform other forms of financial business such as insurance, merchant banking and corporate financial advice with the prior sanction of the committee. However, stockbrokers could not become members of any other stock exchange. Further, members were not allowed to advertise or canvass for or otherwise solicit any work forming part of the business as a stockbroker. As agents, stockbrokers' income was defined by charges set by the Stock Exchange as indicated in Table 2. The Exchange lay down minimum rates and any member charging less than the foregoing rate was liable to a fine not exceeding Kshs 2,000, or an expulsion or suspension or both as the committee may decide.

2.3 Stage III (1963-1970): Political Environment and Kenyanisation Policy

The period immediately after independence saw the Government adopt the Kenyanisation policy with a primary goal of transfering economic and social control to citizens by ensuring that majority of businesses were in the hands of citizens except where some overriding national advantage was otherwise demonstrated. Kenyanisation of businesses

Table 1: Entry requirements, entrance fee and subscription fee for members and registered agents

		1954	1989	1997
Member	Entrance fee (Ksh) Annual subscription fee	2,500 4,000	50,000 6,000	400,000 50,000
	Deposit	50,000 paid back with no interest after probation period	200,000 paid back with interest 3% below saving rate	
	Probation period	At least one year	At least 3 years	
	Working experience	At least three years	At least 3 years	
Registered agents	Entrance fee (Ksh)	1,000	2,500	2,500
	Annual subscription fee	800	1,000	1,000
Securities' representatives	Entrance fee for financial			
	institutions			100,000
	Entrance fee for others			50,000
	Annual fee for financial			
	institutions			20,000
	Annual fee for others			10,000

Source: Rules and Regulations of NSE (1954) and Rules and Regulations of NSE (1997)

Table 2: Commission rates for stockbrockers

	1954		1989		1995
On company	Kshs c	ts	Kshs	cts	1st Kshs 10
shares and	<1.5	0	< 1.5	15	million-1.70
stocks	1.5-2.5	5	1.5-2.5	20	next 40
	2.5-5 2	0	2.5-5	25	Million – 1.50
	5-7.5 2	5	5-7.5	30	above
	7.5-10	30	7.5-10	35	50Million – 80
	10-15	35	10-15	45	
	15-20	10	15-20	55	
	Over 20	2%	over 20	2.5%	
	consideration	n	considerat		
On government	Upto 200,000	-1.5%	Upto 200,000 – 1.5%		Upto Kshs 1
and municipal	Over 200,000 – 1.255		Over 200,0	00 - 1.255	Million – 1%
stocks					Over 1 Million-
					negotiable
Minimum	Below Kshs		Any transaction Kshs		Kshs 100
commission	200 – membe	er	100	_	
	discretion			member	
	200-600 – Ksl		discretion		
	Over 600 – K		*/ 1		** 1.00
Probate and	20 shs per ite	em,	Kshs 30 per item		Ksh 30 per item
portfolio	minimum	10.6	minimum		with minimum
		charge of Kshs 40 for		every	charge of Kshs 100
D () 1	every valuat		valuation	,	for every valuation
Professional	Minimum Ks	shs 400	Kshs 3000	per hour	
advice fee	per hour				F (C 1
					Fee for fixed
					securities 1st 50m-0.0625%
					above 50m – 0.05%
Rebates	A return con		A return co	mmission	above 30III-0.03 /6
Revales	of	1111551011	of 255 allo		
	25% allowed	to			
	registered	to	registered business d		
	agents on bu	siness	their dient		
	done for	511 K 55	discretion		
	their dients	at	members	or tric	
	discretion of		members		
	members				
Short-dated	With one year	r or less	With one y	ear or less	
stock	to		to maturity		
	maturity on i	nominal	nominal ar		
	amount 1%				
	When comm	ission			
Takeovers. new	exceeds				
		n a			
	Kshs 1 millio				
Takeovers, new issues, rights	Kshs 1 millio member at h				
	member at h	is			
	member at h	is ay			
	member at h	is ay quarters			

Source: Rules and regulations of NSE (1954, 1997)

involved transfer of existing firms to citizens and the creation of new enterprises in the hands of citizens. Foreigners held majority interest in companies if sufficient capital was not available from domestic sources or so long as other advantages to the country, such as technology and skills, could only be obtained this way. This was achieved through trade licensing legislation under which lists of businesses owned by non-Kenyans and targeted for transfer to Kenyans by sale within a specified period were published periodically. Kenyans able to take over such businesses were provided with loan assistance by the Government. Therefore, the Kenyanisation policy saw a change in the ownership structure of various businesses. The previously foreign-dominated market saw an increased share of locally-controlled firms.

In addition, to protect the interest of foreign investors in Kenya, the Government passed the Foreign Investment Protection Act (1964), where any proposed foreign investment contributing significantly to development got approved status certificate and owners were allowed to repatriate earnings and capital whenever they wanted. This Act was not nullified by the introduction of exchange control over sterling transactions in 1965 where the latter saw holding back of country money that was previously moving to build assets abroad.

For sometime, the NSE operated as a regional market in East Africa where public sector securities included issues by the governments of Tanzania and Uganda (the East Africa Community). In addition, a few of the listed industrial shares were those of companies registered in Tanzania and Uganda. This was facilitated by maintenance of a common market with free movement of capital, and maintenance of common exchange regulations regarding capital movements to countries outside East Africa (Aworolo, 1971). However, with the changing political regimes among East African Community members, there was need to formalize arrangements governing the relations among members in the East Africa Common Market. A first attempt was in 1964 with the

Kampala Agreement but little was achieved. The East Africa Community Treaty was signed on 1 December 1967 and covered trading arrangements not implemented with the Kampala Agreement of 1964. However, the nationalization measures undertaken by Tanzania in February 1967 and the unilateral decision by Uganda and Tanzania to be excluded from the Scheduled Territories for Exchange Control created uncertainties.⁴ As a result, transactions between Kenya and the two countries were subjected to the normal exchange control regulation, which curtailed free movement and squeezed the scope of the stock market.⁵ It also meant delisting of companies listed in the Nairobi Stock Exchange and originating from Uganda and Tanzania.

2.4 Stage IV (1971-1989): Capital Issue Committee and Taxation Policy

In this period, the Government made a first attempt to directly monitor the operations of the stock market in an effort to ensure that capital raised in the market was not used for investment outside the country. Tight taxation policies were implemented to reduce repatriation of funds by foreigners and to raise Government revenue. However, it was felt that taxation policy changes adopted in the 1970s discriminated the share earnings against other financial assets and also heightened the

⁴ The Scheduled Territories of Exchange Control consisted of Kenya, Tanzania and Uganda as members of the East African Community. It was based on an arrangement that currencies of the three countries would be exchanged freely and at par within these countries.

⁵ The Economic Survey (1968) for example indicates that for a short period, Tanzania government imposed exchange control restrictions against Kenya and Uganda which meant evolving new procedures for making and receiving payments. To restrict capital flight, the Bank of Tanzania suspended the arrangement where Tanzania notes were not redeemed in Kenya and Uganda. With nationalization, Kenyan banks found themselves without correspondent in Tanzania and therefore with no channel of remitting funds or sending cheques and bills for collection in Tanzania.

barriers of listing new issues. For example, dividend income was double-taxed and because dividends were not deductible expenses, corporate earnings distributed as dividends had a high effective tax rate. This made it more expensive for corporations to provide adequate after-tax-return on equity and therefore the preference for debt financing. In addition, corporations were not allowed to deduct expenses of raising share capital of debenture issues from taxable income. As such, the high expenses of raising finance and the non-deductibility of these expenses were clear disincentives for raising equity through public issues. The effective corporate tax was also high and tended to significantly reduce the returns on shareholders.

The Government made a first attempt to regulate the stock market with the establishment of the Capital Issue Committee (CIC) in 1971. The idea was aroused from the observed practice of foreign investors in the process of Kenyanisation. In the Development Plan (1969/70), it is noted that foreign companies selling shares to local investors were following up with the request to remit the funds realized overseas. The Government did not encourage such a move unless it was intended to result to full local control of the company or the investment had the approval status. Therefore, in the Development Plan (1970/74) the Government indicated its intention to introduce legislation aimed at safeguarding the interests of the investing public. The legislation was to oversee licensing of all stockbrokers and creating of a deposit or guarantee acceptable to the Government by a stockbroker with Treasury. This was to be used for compensating investors if the stockbrokers failed to meet their obligation. It was also expected to establish a committee representating the Government and stockbrokers to control and approve new issues of company shares to the public. The proposed committee was to operate as a capital-issue committee for screening issues before they are presented to the public. The screening exercise was to involve receiving details on how funds raised by the issue would be used and

the amount the company wished to repatriate after the issue had been completed. It was also to ask for all information necessary to determine the order of priority for such company issues on NSE. The committee was also to deal with applications from non-resident owned companies wishing to have more than the present allowed credit facilities. In the 1971/72 Budget speech, the Minister for Finance insisted that the NSE should be used by local companies to raise cash for capital development but the Government was not ready to allow the limited local capital to be used to finance repatriation of capital. This, it was felt, would not result in any long-term benefits but a significant loss of short-term balance of payment. However, the Government welcomed the idea of Kenyan capital going into partnership with foreign capital to finance further development of enterprises in Kenya. Consequently, the CIC was set up in 1971 under the Ministry of Finance, consisting of representatives from the Treasury, the Ministry of Commerce and Industry and the Central Bank of Kenya. The key responsibility of the committee was to advise the Ministry of Finance on terms, priority and timing of all new public issues of equity, rights issues to existing shareholders in publicly-traded companies, debentures and loan issues. It also had the responsibility of approving the price of issues, timing of sales and allotment of the issues between institutions and individuals, and also between residents and non-residents. However, there was no clear criteria on which decisions were made. Lack of full information had a great impact on the ability of firms to raise capital publicly.

Further, to control movement of funds outside the country, a tight tax policy was adopted. In 1971 for example, the Government imposed a 12.5% tax on dividends and interest paid by corporations to residents and non-residents. In addition, a 20% tax on all fees and royalties payable to non-residents was imposed. The capital gain tax was aimed at generating additional revenue and curb excessive speculation. Further,

in the 1974/75 fiscal year, the rate of tax on dividend paid to nonresidents was raised from 12.5% to 15%; the rate of withholding tax on resident dividends also went up to 15%. To encourage the inflow of capital, the Central Bank in 1974 allowed foreign-controlled companies to turn more to overseas markets for financial requirements. Local borrowing was restricted to 20% of investment for a few exceptions. To ensure compliance, the Central Bank withdrew permission for foreign firms to remit interim dividends or profits prior to the finalization of the annual accounts of their firms. The Bank also withdrew the general permission for local borrowing by firms whose non-resident ownership exceeded 15%. The withholding tax paid by non-residents was raised from 15% to 20% and interest rate from 12.5% to 20%. Stamp duty was raised to 3% in the 1973/74 period while withholding tax for dividends was raised to 15% in 1974/75. In June 1975, a capital gain tax of 36% was introduced. This was progressively reduced by 50% in 1981 and 25% in 1982.6 To boost growth of the stock market, there was need to put in place tax incentives, including suspension/elimination of capital gain taxes, withholding taxes on dividends and making public offer expenses tax-deductible. Therefore, in the 1985/86 fiscal year, capital gain tax was suspended on the ground that tax inhibits capital and share mobility and therefore economic growth.

During the period, there was no major change in the composition/range of securities issued in the stock market. However, there was a rise in the proportion of Government stocks listed in the NSE. The ratio increased from 5.4% in 1954 to 13% in 1959, while in 1969 all Government

-

 $^{^6}$ The 1981/82 adjustments made on the capital gain tax saw only 50% of the net gains subjected to corporation tax such that with a corporation tax of 45%, only 22.5% of net gains was paid as tax by company. For individuals, tax on gain was progressive for 10% up to 35%. In addition, withholding tax on investment shares was reduced from 35% to 15% aiming to boost stock exchange (IFC/CBK, 1984).

securities were listed in the NSE. This followed the takeover by the Central Bank of the administration of the local register of Government stocks in July 1968, including the Nairobi register of stocks issued in London stock market. By 1968, the number of listed public sector stocks was 66 of which 45% were for Kenya government, 23% Tanzania government and 11% Uganda government. By 1969, the amount of London-issued stocks on the Nairobi register was Kshs 3 million compared to Kshs 1.4 million in 1963, indicating a shift of Kshs 1.8 million of which Kshs 0.8 million was due to imposition of exchange controls against sterling area in 1965.

Government stocks attracted very low returns and were unpopular in the private sector. For example, in 1954 they attracted 2.5-3.0% interest rates, increasing to 6.5% in 1965 and remaining almost constant at 6% between 1967 and 1973. The period of maturity increased overtime where it ranged between 5 and 10 years in 1965, 6 and 18 years in 1967 and was over 20 years in 1970, showing a shift to long-term funding. Most of the stocks were taken up by agencies controlled by the Government and those directed by the Government to invest in these securities. For example, there was an increase in holding of Government stocks by the insurance companies in 1965. This followed Government policy of encouraging more insurance funds to be invested locally, therefore covering local liabilities with local assets. This was also enhanced by the introduction of exchange control which prohibited the transfer of life insurance premium abroad (Government of Kenya, Economic Survey, 1969). However, the National Social Security Fund (NSSF) which was established in 1966 became a major holder of Government stocks and dominated the market from 1969.

Despite the efforts made to promote the EAC, Uganda imposed exchange controls on Tanzania and Kenya in 1970. This meant that payment to residents required authorization from the Bank of Uganda and Uganda notes were not to be exported. This hindered intra-trade

in East Africa and when the EAC finally collapsed in 1975, the Uganda government compulsorily nationalized companies which were either quoted or subsidiaries of listed companies.

3. Revitalization Process

Despite the efforts made to promote growth of the capital market and the financial sector in general, the contribution of the sector to economic development was viewed as unsatisfactory as the economy hanged on the balance with dwindling inflow of foreign savings. Consequently, the Government viewed the reform of the sector as the best option especially if the economy was to shift its reliance on domestic resources to finance domestic investment (Government of Kenya, 1974/78). Despite the Government realizing that the capital market was playing an insignificant role in the development process in the early 1970s, as characterized by weak development of financial institutions and a thin and inactive stock market, revitalization reforms were only implemented in the 1990s following the IFC/CBK (1984) study whose recommendations became a blueprint in the reform process. The Government reaffirmed its commitment to the reform process in Sessional Paper No. 1 of 1986 (GoK, 1986). The recommendations emphasised the need to develop money and capital markets by diversifying money market instruments and removing taxation differences between the debt and equity finance in order to achieve diversity in the sector. Another proposal made was the need to establish a regulatory authority with the powers to provide regulatory measures for improvement and proper functioning of a fair and orderly market. The reform process was implemented simultaneously with the overall financial sector reforms aimed at enhancing efficiency in the price discovery process, reducing transaction costs, facilitating competitiveness and increasing liquidity. A statutory regulatory framework was adopted during the reform period charged with the responsibility of protecting the interests of investors in addition to developing the stock market.

3.1 Statutory Regulatory Framework

A statutory regulatory framework was established as part of the ongoing capital market reforms in an effort to strengthen the regulatory infrastructure.⁷ The establishment was recommended by the IFC/ Central Bank of Kenya study (1984) and reaffirmed in the Sessional Paper No. 1 of 1986. In order to establish a regulatory body, a legislation was adopted to facilitate formulation of rules and enhancing the effectiveness and efficiency of the operations of the Capital Market Authority (CMA). The Authority was established in 1990 through the Capital Market Authority Act (Cap 485A). The process involved the Government setting up a Capital Market Development Advisory Council in November 1988 whose role was to work out modalities for establishing the CMA. In November 1989, the Bill was passed by Parliament and the CMA was constituted in January 1990 and inaugurated in March 1990. It is important to note that while statutory regulation is desirable to provide necessary oversight of the market and ensure its stability by inter alia preventing conflict of interest, it should complement rather than substitute self-regulation (Akamiokhor, 1996).

The main purpose of setting up the CMA was to have a body specifically charged with the responsibility of promoting and facilitating the

⁷ In developed stock markets, a similar trend exists where securities' regulatory bodies are set up mainly to curb specific abuses that occurred in an already developed market and not necessary with the primary aim of developing the market. The London Stock Exchange, for example, established a market regulator with the Big Bang in 1986 after being operational since 1773 (Akamiokhor, 1996). The argument was that the self-regulatory framework was designed to protect the interest of members and could not at the same time protect the public against its members. The NYSE established a SEC with an aim of restoring confidence among investors after the stock market faced crises in 1929. Market regulation is geared towards protecting the integrity of the market place, member firms and the customer, with more emphasis on protection of investors, ensuring fairness, market integrity and minimisation of systemic risk.

development of an orderly and efficient capital market in Kenya. 8 As spelt out in the Capital Markets Authority Act (Cap 485A, 2(11)) the aim of the CMA is to develop all aspects of the capital market with particular emphasis on the removal of impediments and the creation of incentives for long-term investment in productive enterprises. It also targets to protect investors' interests by operating a comprehensive fund to cushion investors from financial loss arising from the failure of a licensed broker or dealer to meet contractual obligations.

The CMA is composed of eleven members who include: the Chairman, who is a non-executive member appointed by the President on recommendation of the Minister for Finance; six members who include professionals in law, business, accountancy, finance and insurance and appointed by the Minister for Finance; the Chief Executive who is appointed by the Minister for Finance; and three permanent members, that is the Permanent Secretary (Treasury), the Governor of the Central Bank of Kenya, and the Attorney General. Except for the three permanent members, the rest are appointed for a three-year period and are eligible for re-appointment.

_

⁸ Established regulatory bodies in emerging markets are charged with the responsibility of regulating and developing the stock market. In developing the market, regulatory authorities are charged with the responsibility of improving the regulatory environment in order to provide adequate return on equity investment and assuring its comparability with other savings' instruments. Such bodies are also responsible for ensuring sustainability of market liquidity by enhancing supply and demand for securities and also encouraging the development of securities' intermediaries such as stockbrokers and underwriters/merchant banks. The authorities are also charged with the responsibility of enhancing diversity of financial assets by encouraging issue of a broader range of securities. In regulating the market, the authority aims to ensure proper functioning of a fair and orderly market with tight disclosure rules for both corporate and market information; adequate organization and self-regulatory duties of the exchange, registration of securities' professionals and protection of investors. While the development task is taken as an overriding objective, the regulatory role is implemented when it is demonstrably necessary and if it is not going to affect the development of the market. It is however important that the rules made do not over or underregulate the market, but should be targeted to encourage innovativeness and competitiveness and maintain high standards of business conduct.

Efforts were made to make the regulatory body financially independent. As indicated in the CMA annual report (1990/91 and 1991/92), the main source of revenue for CMA when it was established was limited to licensing fees of market players and this was not adequate to meet its operational costs. The Government was therefore financing a significant proportion of the CMA activities, which was supplemented by substantial donor support from the United States Agency for International Development (USAID). However, there was growing need to make CMA self-financing and reduce its reliance on the Treasury. In the 1995 CMA (Amendments) Act, the Authority was allowed to collect fees from the market to finance its operations. A framework to facilitate the charge of fee for its services was developed and in January 1996, the CMA was allowed to levy charges on public issues, bonuses, rights and private placement by public companies. The new fee structure was gazetted in July 1997 empowering the CMA to charge fees for its services mainly from public issue stock exchange transactions and licensing. At the same time, the Government in 1996 exempted the Authority from the State Corporations Act (Cap 446 of the Laws of Kenya) to facilitate flexibility and enhance autonomy in discharge of its operations. This saw the dependencies on the exchequer reduce from 100% to less than 28% by end of June 1998. The CMA took a zero-based budgetary approach which required each budget item to be reviewed and justified afresh every financial year regardless of the position of expenditure or revenue in the previous years (CMA Annual Report, 1997/98). The charges, however, seem to have substituted the taxation relieves for both the listed companies and investors. As a result, the effective withholding tax was more than the set 7.5% while the listing costs were still high though tax-deductible for firms intending to go public.

To meet its developmental role, the CMA (Amendments) Act of 1995 spelt out regulations providing for foreign investors' participation in

the domestic capital market. This facilitated the expansion of market scope. In 1997, guidelines to govern the issuance of corporate bonds and commercial paper were issued in order to facilitate diversity of financial assets traded in the market. To facilitate its regulatory role, the 1996 review spelt out the requirements on take-over, mergers and de-listing to protect minority shareholders. The CMA also issued guidelines to promote good corporate governance practices by listed companies through the constitution of audit committees in January 1999. Further, in July 1995, the CMA established the investor compensation fund aimed to compensate traders for financial losses arising from the failure of a licensed broker or dealer to meet contractual obligations.⁹

In 1998, the CMA published new guidelines on the disclosure standards by listed companies which required them to form audit committees composed of non-executive directors with knowledge of business and accountancy. The disclosure requirements were meant for both public offerings of securities as well as continued reporting obligations, among others.

3.2 Trading System

In November 1991, share trading moved from coffee-house to floor-based open outcry system. The open outcry system was opted for to

_

⁹ The compensation fund is financed as follows. First, every licensed stockbroker is required by law to maintain with the Stock Exchange an irrevocable bank guarantee by a commercial bank for Kshs 1.5 million which could be used to settle any inter-broker or third party claims arising from day to day unsettled trades. Second, the NSE is required to maintain another level of compensation which is paid by buyers and sellers of securities at a rate of 0.01% of the consideration. The fund could be used when bank guarantee is not adequate. Third, the CMA (Amendment) regulation of 1995 requires every buyer/seller of securities to pay 0.01% of the consideration to the investor compensation fund. This fund is accessible if the first and second compensations are fully utilized and a legal proceeding bankruptcy of liquidation proceedings against the licensed person are complete and net losses determined.

enhance transparency by according all brokers an equal opportunity to bid for securities and also to enhance handling of the growing trading activity. This followed the recommendations made by the IFC/CBK (1984) study justifying that since brokerage businesses were conducted behind closed doors, the stock exchange had not been able to generate adequate public awareness and confidence in the buying and selling of securities. The trading system, it was felt, did not guarantee that the prices obtained by buyers and sellers are best since all buying and selling interests did not get exposed to one another.

With the new trading system, trading took place on Mondays to Fridays between 10.00 a.m. and 12.00 noon except for public holidays or any other closures approved by the Board of Directors of the NSE. Potential buyers and sellers contacted stockbrokers directly or through registered agents and gave their buying or selling orders. Foreign institutional investors were supposed to have a current custodian account with a Kenyan banking institution. Then, stockbrokers through their authorized representatives took the customers' orders to the trading floor where deals were transacted using open outcry auction. Trading board lot was in terms of one hundred shares or units while trading was effected to and from the trading board on the trading floor and no private negotiation was permitted. Only when bid price was equal to or up to two spreads away from the offer price is transaction effected. There were daily limits on movements of quotations whether bid or offer set at 15% of opening bid or offer prices. No bid or offer quotations were more than six spreads from the last quotation appearing on the trading board for that security (NSE, 1997b). The following were the bid spreads set for shares, stocks, rights, options and other securities: for less than Ksh 20 the spread was 5 cents; Ksh 20-Ksh 50 was 25 cents; Ksh 50 -Ksh 100 was 50 cents; and over Ksh 100 was 100 cents.

The manual system of clearing and settlement lengthens the financial transaction period. As noted in the NSE Annual Report (1998), excluding

any unforeseen bottlenecks, transfers are usually affected within a week or two following transactions. Such a long period implies that investors waste a lot of time in between the actual sale and its confirmation, which could adversely affect the liquidity and efficiency of the stock market. Lack of a central depository system was therefore seen as inhibiting the liquidity of the stock market.

A proposal was made to install the Central Depositing System (CDS) in 1995 following a feasibility study which identified the urgent need for CDS as a critical factor among three critical issues. The other two issues included the need to establish an enabling legislation to ensure best practice and the need to acquire the best technology. A draft Bill was submitted to the Attorney General for finalization in March 1996. Installation of a CDS was aimed at enhancing liquidity and efficiency in the trading system by reducing the period of delivery and settlement. This, it was hoped, would facilitate electronic transfer of ownership without the physical movement of such certificates, minimizing systemic risk and the settlement period and ultimately increasing trading volume. The main aim of the CDS was to shorten the registration process, boost liquidity in the market, increase market activity, reduce market risk, attain international standards and deliver the NSE mission statement.

Three years down the line, little had been achieved. The newly appointed Executive Director of NSE attributed the delayed implementation to lack of proper CDS project planning. To this effect, the CDS project was launched with a detailed timetable of events. A decision was made to lay down the infrastructure of a CDS company while awaiting the passing of the Bill. The NSE also engaged in lobbying Members of Parliament to pass the CDS Bill promptly when it is presented before them. In laying the infrastructure for a CDS company, it was agreed that shareholders of the company would pay 1% of capital upon signing of the agreement and 10% upon the Bill obtaining Presidential assent, while the balance would be called up by the directors

as the need arises. The shareholders included: the NSE, Citibank, Kenya Commercial Bank, Apollo Insurance, IFC Jubilee Insurance, stockbrokers and the Cooperative Bank (which was showing interest). On 7 March 2000, five core shareholders signed an agreement and paid up 1% of the share capital. The Central Depository and Settlement Corporation Limited (CDSC) was incorporated under the Companies Act (Cap 486) on 23 March 1999. The Bill passed its seconding reading in Parliament in April 2000 and the draft CDS Bill was tabled in Parliament in December 1999. However, Parliament went on recess before debating the Bill. In June 2000, registrars, custodians and stockbrokers hosted by the NSE management held a workshop to brainstorm on the best practice in the interim period prior to implementation of the CDS. The aim was to create an even understanding by all market players of the new changes and benefits thereof and prepare them for the high pace needed in the automation process. The request for proposals for CDS and ATS was then floated on 20 June 2000. On 1 July 2000, the chief executive of the NSE made a presentation on the proposed new system of delivery, settlement and registration of securities as a precursor to the implementation of the CDS. Bids from 11 prospective suppliers of CDS and electronic trading system were opened on 17 July 2000. This was followed by a workshop for stockbrokers, custodians, registrars, and back-office personnel aimed at enhancing the goodwill of all stakeholders with formal awareness campaign. Further, the immobilization of over half a million share certificates held by investors so as to propel the department from a semi-computerized division in settlement cycle to a settlement and registration company for all listed securities was launched. This saw the introduction of delivery versus payment system (DvP) on 1 August 2000. With this system, the market faced a main challenge of settling transactions within 5 days of trading occurring and providing shareholders with their shares within 7 days of settlement, creating an environment of a smooth transition to an electronic based settlement and registry. The main aim of the DvP system was to enhance investor confidence and liquidity by making the settlement period shorter and safe. It was also aimed at moving the market close to a CDS environment of T+3 by shortening the period within which certificates are issued and centralizing registration at the NSE. The system was also aimed at enabling brokers to concentrate on core business by taking over post-trading back-office tasks at the NSE. The CDS Act was passed by Parliament and received Presidential assent paving way to implementation of the CDS on 18 August 2000. Opening of bid proposals submitted by the four shortlisted suppliers of CDS and ATS followed on 25 September 2000. However, there were further technical complications that saw the process delayed.

3.3 NSE Membership

The NSE was registered as a limited company under the Companys Act in 1991. With the 1994 CMA Act (Amendments), it became mandatory that a stock exchange approved by the CMA was to be a company limited by guarantee. The NSE Board was expanded from the original five members to incorporate the interests of other market players in the decision-making process. The new Board comprised of 5 brokers drawn from the main brokerage firms, 2 members from listed companies and 3 members to represent investors' interests. In the NSE rules and regulations of 1997, it is observed that no dealer or stockbroker would be granted membership unless they possess a license to operate as a broker or dealer. Listing of companies saw change in requirements and procedures over time. Before the reform period, for example, firms were required to have a minimum of Kshs 2.9 million issued and paidup capital and to make a public offer of not less than 20% of the authorized capital, or shares with a nominal value of Kshs 1 million, whichever was less.

In July 1994, stock brokerage firms were increased for the first time with licensing of seven more stockbrokers. An additional eight stockbrokers were licensed in June 1995 and with the suspension of one stockbroker, the total number of stockbrokers was twenty. The expansion of brokerage firms was meant to enhance competitiveness in trading. This followed the peaking of stock price in February 1994 to 5030, which may have created an incentive to share in the profits. In March 1997, stockbrokers formed the Association of Kenya Stockbrokers (AKS) with the objective of developing a code of conduct, promote professionalism, and establish examinable courses for its members as well as facilitate liaison with the CMA and the NSE. In 1993, the CMA increased the initial paid up capital for stockbrokers from Kshs 100,000 to Kshs 5 million while that for investment advisors was set at Kshs 1 million.

In addition, there were considerations to encourage participation of foreign brokers and fund management companies in the domestic capital markets through joint ventures. This proposal was subject to a minimum of 51% local shareholding for brokerage firms and 30% for fund management companies. The idea was to facilitate the transfer of skills and technology into the local market, encourage portfolio investment into the domestic market, and help position Kenya globally within the competitive financial markets (CMA Annual Report, 1997/98). There were proposals to establish merchant banks to promote underwriting services and a rating agency. This was motivated by the fact that as markets became more sophisticated in terms of range of products and activities, demand for accurate information and risk assessment by investors increased in order to provide objective analysis and rating of listed securities.

There were also plans to license market dealers at the stock exchange as part of measures to enhance liquidity. The CMA Amendment Act 1994 provided for licensing of dealers; one dealer was licensed in 1999.

The dealer applied to be an investment bank in May 2002 without taking deposits but to provide investment advisory services. Another stockbrokerage firm, Hak securities, was allowed to operate as an investment bank after fulfilling the requirements. Licensing of investment advisors was passed in 1994/95 and nine investment advisors were licensed to provide vital services to investors as professional analysts and portfolio managers. By 1997, the number of investment advisors had increased to 19.

In addition, in the 1997/98 period, Foreign Fund Managers and Investment Advisors were allowed to participate in the local capital markets by acquiring beneficial ownership in local stockbroking firms and fund managers; precisely up to 49% in the case of brokers and up to 30% for fund management companies. The aim was to induce specialized expertise in fund management and foreign investible funds to start flowing into local markets.

3.4 Foreign Investors

During the revitalization period, capital controls were relaxed for offshore borrowing in February 1994 subject to quantitative limits. Complete liberalization of offshore borrowing was implemented in May 1994, while some restrictions on inward portfolio investment were lifted in January 1995, therefore allowing participation of foreign investors in the Nairobi Stock Exchange trading under guided policy.

The following are the rules set for foreign portfolio investors in the Nairobi Stock Exchange as contained in the Kenya Gazette Supplement of January 1995:

 No foreign security (capital market instrument) which is not listed in the securities' exchange in Kenya shall be offered for sale or exchange without the approval of the CMA.

- No person shall offer for sale any capital market instrument registered outside Kenya except with the approval of the CMA.
- A listed company should maintain a register of shareholders of ordinary shares showing the amount of ownership by individual citizens of Kenya, institutions registered in Kenya, and foreign investors.
- A foreign-controlled company shall not be eligible for further investment by foreigners other than through a new issue of shares.
- Any issuer making an issue of shares to the public (primary market) may offer part of the shares for subscription by foreign investors but this should not exceed 20% of the issued share capital.
- Every capital market instrument issued outside Kenya and approved by the CMA for sale in Kenya other than unit trusts and mutual funds licensed by the Authority shall be listed with a securities exchange in Kenya.
- Every share certificate registered in the name of a foreign investor shall be held by an authorised depository.

Foreign investors were permitted up to 20% of equity for inward portfolio investment, that is aggregate of each stock and a 2.5% limit for individual investors. This was revised upwards in July 1995 so that the individual investor limit was increased to 5% while the aggregate was set at 40%. On 26 July 2002, new foreign investor regulations were established where a 25% minimum reserve of the issued share capital was for locals while the balance of the 75% was a free float for all classes of investors. The 25% minimum reserve applies during initial public offerings (IPOs) and Government of Kenya privatizations.

Foreign investors' trading recorded a peak of 52.5% of the total market trading in December 1996 while a consistent downward trend was recorded from June 1997. Initially, when foreign investors were allowed to trade in January 1995, a slow growth in foreign trading was recorded. However, inflow of foreign trade increased tremendously with the increased limit of foreign participation in July 1995. There was tremendous growth in foreign investors' activities, increasing from 3% in 1995 to 44% in June 1997, with a peak in December 1996 of 52.5%. It is therefore tempting to conclude that market performance during the 1996-1997 period was to a large extent controlled by foreign operations.

The remarkable performance of foreign trade was short-lived as the ratio to total trade declined to 2% in August 1997 before recording a net-outflow of 42% in September 1998. Across the sectors, the agricultural sector lost foreign trading especially in 1998, and this was coupled with a decline in performance of the sector. The financial sector seemed to take the remnant of the foreign trading although growth of the sector took a downward trend. Distribution of shareholders between locals and foreigners indicated that the majority of shareholders were locals (in a sample of 38% of the listed companies). The skewed distribution in shareholding was indicated across the listed companies where the top 20 shareholders, representing 6% of the total shareholders, held more than 50% of the shares (from a sample of 57% of the listed companies analyzed). The trading volume results show that the activities

-

¹⁰ The impact of foreign investors on the performance of the stock market is however not clear. It is for example observed that greater foreign participation in the market reduces price volatility. De Santis and Imrohoroglus (1997) explain that new investors broaden the market, dampening the effect of order flow shocks on prices and may also make prices efficient by increasing the precision of public information regarding fundamental values. Sellin (1996), however, regards foreign investors' participation as noise trading, therefore a source of excess volatility in the market. Moreover the general literature on capital flows indicates that when capital flows are push-factor driven then they are more volatile.

of the secondary market were however dominated by the minority shareholders. For example, the average number of shares per deal was estimated at 2,452 shares in 1998, a drop from 3,293 in 1997, while the average value of the deals was Kshs 162,385 in 1998 and Kshs 195,551 in 1997. It would therefore seem that the top shareholders controlling over 50% of company shares are inactive in the secondary market, showing preference for dividend income to capital gains. This may explain the low liquidity in the secondary market and low supply of securities for trading.

3.5 New Listings

With the reform process, the paid-up capital increased to Kshs 20 million in 1997 and then to Kshs 50 million in 2000 as indicated in Table 3. However, to reduce the listing costs, the Government provided various tax incentives where, for example, costs of initial public offerings for shares, debentures and bonds were made tax-deductible expenses; this was announced during the 1991/92 fiscal year. Further, during the 1997/98 budget speech, all the costs incurred by companies in the process of acquiring international credit rating were made tax-deductible. The aim was to enable firms' access to cheaper funds from foreign capital markets by for instance floating debt securities offshore. However, part of these benefits were crowded out by the costs charged by the CMA on market participants; this, together with other listing requirements, failed to significantly cut down the entry barrier for new companies.

The procedure for company listing changed as the Government increased its involvement in market activities. For example, while before 1971 a firm needed only the approval of the NSE, the listing firm now had in addition to receive the approval of the committee in order to list in the NSE. With the review of the *Rules and regulations of NSE 1954* in 1989, a listing company had in addition to notify the NSE on any

Table 3: Financial obligations for new listing

	Minimum issued and fully paid canifal	Amount of issued capital	Hearing fee	Fees	A ddition quotation	A mual quotation fee	
1954	At least Ksh 2m and shares to have a nominal value of Ksh 5	Amount issued not <20% of authorized capital or shares to nominal value of Ksh 1m, whichever is less	Ksh 1000 payable for the 1st issue when application is submitted by the sponsoring broker	Ksh 1000 for each class of shares or debentures paid to exchange within 30 days of quotation being granted.	Ksh 1000	Less Ksh Im – 1000 Im to 2m – 1400 2m to 3m – 1800 3m to 4m – 2200 4m to 5m – 2600 5m to 7.5m – 3200 7.5m to 10m – 4000 Over 10m – 6000	
1989	Ksh 2m	Not <20% or nominal value of Ksh 1 million whichever is less	0.1% of the issue nominal value of securities to be listed subject to a minimum fee of Ksh 50,000 and max. fee of Ksh 500,000	0.025% of the value of each class or debentures minimum Ksh 3000	0.05% of the issue nominal value of securities to be listed subject to a min. fee of Ksh 25, 000 and max. fee of Ksh 100,000	0.01% of the total market value of the quoted securities subject to a minimum fee of Ksh 3,0000	
1997	Ksh 20m	Not <20% of its paid-up share capital held by not <200 shareholders.	0.2% of the value of securities to be listed minimum Ksh 200,000 and maximum Ksh 25 million		0.1% of the market value minimum Ksh 100,000 and maximum 1 million	0.05% of the market capitalization of listed securities minimum Ksh 200,000 and maximum 2 million	Offer for sale fees payable by a holder of large block of securities in a listed company offering to sell the block to public investors. 0.15% of the market value of the securities being sold subject to a min. 20,000 and max. Im
2000 MIMS	Ksh50m		0.1% of nominal value of additional securities to be listed subject to min. Ksh 50,000 and max. 500.000	Initial fee of 0.06% of value of securities to be listed subject to a min. of Ksh 200,000 and max. of Ksh1.5m			
AIMS	Ksh 20m		0.1% of nominal value of additional securities to be listed subject to min. Ksh 25,000 and max. 250,000	Initial fee of 0.06% of value of securities to be listed subject to a min. of Ksh 100,000 and max. of Ksh			

information necessary to enable the shareholder and public to appraise the position of the company and to avoid the establishment of a false market in its securities. Listed firms also had to announce through the sponsoring broker any mergers and take-overs when negotiations are at an advanced stage and apply for a temporary halt of dealings if this may lead to a very damaging sharp movement in share prices.

The NSE has also witnessed changes especially following the reform strategy launched on 8 May 2000 by the CMA for reorganizing the stock market. The reforms were aimed at responding to the changing needs of the market and the economy as well as deepening the capital market. The restructuring of the market was aimed at catering for the specific needs of different issuers. Three segments were established and launched to the NSE Board on 5 February 2001. The three segments include the Main Investment Markets (MIMs), the Alternative Investment Markets (AIMs) which targets the small and medium-sized companies with potential to grow, and the Fixed Income Securities' Markets (FISMs).

3.6 Regional Market

There are presently on-going efforts towards integrating and harmonizing the operations of the East Africa economies. A lot of effort has been put in establishing an appropriate infrastructure that would facilitate the establishment of an East African Regional Capital Market. This would harmonize the legal and regulatory framework, trading system, and the taxation policy. It is hoped that a unified East Africa stock exchange will be established to take advantage of the large population estimated at 80 million people. A regional market would facilitate access by companies to a greater pool of resources; ease of

cross-border listing of securities; increase liquidity; and enhance diversity of investors' risk. In addition, a large capital market would attract foreign investors. In March 1997, the East Africa Member State Securities Regulatory Authorities (EASRA) was formed through a memorandum of understanding between the securities' regulators of Kenya, Uganda and Tanzania. While the EASRA initially comprised of the regulatory authorities, the opening of stock markets in Tanzania and Uganda implied that membership was extended to the stock exchanges. Further, during a meeting on 11 August 2000 Ministers of Finance for Kenya, Uganda and Tanzania proposed the establishment of the Capital Market Development Committee (CMDC) as a separate committee under the EAC framework. This was endorsed during the 31 August 2000 meeting of the council of ministers. The main aim is to harmonize policies and proposals of importance to the capital markets of the three EAC countries. Further, the East African Legislative Assembly (EALA) was launched on 30 November 2001 by the three East African heads of state.

3.7 Securities

In an effort to deepen the capital market, corporate bonds were introduced. There was also a move in 1997 for NSE to work towards introduction of collective investment vehicles (unit trusts and mutual funds) and other corporate bonds including revenue bonds, mortgage-backed securities and other viable instruments. The same emphasis is made in the 1999 report where establishment of collective investment vehicles like unit trusts, mutual funds and other formal investment clubs would allow investors' access to a fully diversified investment in a single unit, greatly enhancing their ease of access to the market. Corporate

bonds¹¹ were introduced in the market on 22 November 1996 when the East Africa Development Bank (EADB) bond was issued at a price of 99% to raise Kshs 600 million. It was traded in denominations of Kshs 1 million with an interest of 1.2% points above the prevailing 91-day Treasury bill rate. The bond was redeemed in six semi-annual installments. Coupon interest payments were made quarterly in arrears. The bond can be purchased/sold on the secondary market through the NSE. Corporate bonds were traded in denominations of Kshs 1 million with settlement in T+7 days. The Government targeted to issue three one-year floating bonds and the first issue was made in April 1997 with a subscription of 2.6 billion out of the Kshs 5 billion tendered. Low subscription was attributed to the failure of the Central Bank of Kenya to give sufficient notice for proper placement of the bond. This was a one-year maturity fixed income security with new issues floated quarterly, and with a minimum purchase in primary auction of Kshs 50,000. The nominal annual yield was indexed 2.5 points above the 12 week moving average of the 91-day Treasury bill yield and reset quarterly. Coupon interest payments were made quarterly in arrears. The bonds were purchased/sold on the secondary market through the NSE. These were traded in denominations of Kshs 50,000 with settlement in T+3 days. Interest rates on bonds were made less 15% withholding tax for Kenyan residents and less 12.5% withholding tax for non-

¹¹ Corporate bonds were targeted to provide long-term financing requirement. For a company to issue bond and commercial paper, it had to have a paid up share capital and reserve of not less than Kshs 50 million and would be retained at the level during the period the bond was outstanding. If the minimum paid up share capital was not met, then the issuer would obtain from a bank or any other approved institution a financial guarantee to support the issue. The issuer should have made profit at least two of the last three financial periods preceding the application for the issue. The total indebtedness including the new issue of bond should not exceed 400% of the company networth (gearing ratio 4:1 at the date of the latest balance sheet). The funds from operations to total debt for three trading periods proceeding the issue should be maintained at a weighted average of 40% or more. The minimum size of bond issue is Ksh 50 million and the minimum issuing lots Kshs 100,000-offer period in maximum 10 days, while commercial paper minimum size is Kshs 1 million.

residents. The interest on the bond issues was benchmarked on the 91-day Treasury bill rates. In 1998, the Government issued the first two-year Floating Rate Bonds as part of the objective to shift the short-term debt consisting mainly of Treasury Bills to long-term debt and also to help address the inverse yield curve. Also, in the 1998/99 fiscal year, the Government proposed the issue of tax amnesty bonds to encourage compliance with the Income Tax Act. At the same time, it was made a requirement that insurance companies invest a minimum of 25% of their gross premiums in Government securities, therefore providing liquidity for the bonds market. While the Government bond ranged between one to two years, the EADB bond was made up to three years.

Commercial paper was also explored as an alternative for short-term working capital financing by the corporate sector. However, only those companies listed at the NSE were permitted to issue commercial paper. The guidelines for issuance of commercial paper and corporate bonds were finalized in 1997.

The period 1996/97 saw the launching of investment funds and venture capital funds, including the Regent Undervalued Assets Africa Fund that applied for listing in Kenya, Botswana and Ireland. This offshore investment fund was listed in the NSE in 1997. The fund offered 10 million shares at US\$ 10 out of which Kenya was allocated 10.2% to subscribe for with minimum subscription per investor fixed at US\$ 100,000. Simba Fund was set up by Barring Asset Management as a regional investment fund for Africa with a resource pool of US\$ 120 million. The Fund was invested in seven sub-Saharan African countries with favourable GDP growth rates. The Acacia Fund Limited was the first Kenyan venture capital fund. The Commonwealth Development Corporation with a capitalization of Kshs 1 billion promotes the Fund. The Fund was launched as a ten-year close-ended fund that would make equity investments in medium-sized companies which have the potential of being listed at the NSE. There were also efforts to help

mortgage finance companies to issue asset-backed securities to facilitate increased savings in the financial assets and at the same time develop alternative financial market instruments to raise capital. The main aim was to facilitate the raising of additional long-term capital which would contribute towards the development of the housing sector.

Another factor that might affect the operations of the market is the Retirement Benefits Act (RBA Act). The enactment of the RBA Act and the NSSF Amendment Act of 1997 gave some hope to the stock market. The move to professionalise pension management was seen as crucial in developing domestic sources of long term savings for investment. It was hoped that contracted savings such as insurance and pension funds would spar growth of a vibrant bond market in Kenya and give the stock exchange a tremendous boost. It was expected that statutory institutions which have over-invested in property would readjust their portfolio towards equity in line with the 30% limit proposed by the RBA, while the requirement that all funds be managed by professionals is expected to result to more optimal asset allocation in the sector.

3.8 Taxation Policy

During the reform period, tax policy was relaxed to reduce transaction costs by providing a favourable tax regime that exempts listed securities from stamp duty, capital gain tax and Value Added Tax (CMA Annual Report, 1997/98). Tax reform also saw lowering of withholding tax dividends while the repeal of the Exchange Control Act removed restrictions on transactions for both local and foreign currencies, payments, securities, controls of imports and export, transfer and settlement of property. There were four important developments regarding the taxation policy, namely the 1990/91 policy changes on stamp duty and other transaction costs; the 1996/97 policy changes on withholding tax; and the 1997/98 restructuring of investment and fund management taxation and the 2001/2002 tax incentives for new listings.

In the 1990/91 fiscal year, stamp duty payable for retail share transfer transaction on quoted securities was abolished. In addition, a withholding tax rate of 15% on dividend income paid to residents was made final tax. This was aimed at reducing incidence of double taxation on corporate dividends and incomes to individuals arising from investment in securities. All dividend and interest income to unit trusts were made subject to withholding taxes of 15% and 10% respectively such that this was a final tax not subject to further corporate or personal tax. Further, the withholding tax on dividend income was reduced in 1992/3 to 10% and rationalized with the rate applicable to interest income.

In the 1996/97 fiscal year, withholding tax on dividend income was lowered from 10% to 5% for local investors while it went down to 7.5% for non-residents. Withholding tax on interest income was raised from 10% to 15% while withholding tax on bearer instruments and certificates of deposit was raised from 10% to 20%, and it was the final tax in both cases.

Further fiscal reforms were made in the 1997/98 period. The gains that insurance companies were making from trading at the stock market and which were previously taxable at the normal corporation tax level were exempted from tax. The aim was to release for trading large stockholdings which were in the custody of insurance companies and therefore boost the supply of securities in the market. The measure was also aimed at releasing funds held by insurance companies for investment in listed securities, therefore boosting demand in the market. In addition, sole-purpose dealers who would become tax-exempt were to be licensed with a view to enhancing liquidity in the market. In order to discourage these dealers from locking up securities, the dealers were required to trade them within 24 months from the date of purchase or else suffer taxation. Finally, in order to encourage equity in medium-

sized companies with growth prospects, venture capital funds were allowed to enjoy a 10-year tax holiday on their dividend incomes. 12

In order to encourage new listings, new and expanded share capital by listed companies or those seeking listing was exempted from stamp duty in the budget for the financial year 2000/2001 while expenses incurred by companies in having their financial instruments rated by an independent rating agency were made tax-deductible. Further, in the budget for the financial year 2001/2002, newly listed companies approved under the Capital Markets Act were to be taxed at reduced corporation tax rate of 27% as compared to the standard rate of 30% for three years following the date of listing. However, such companies should offer at least 20% of their share capital to the public. The companies that apply and are listed were to get a tax amnesty on their past omitted profits subject to making a full disclosure of their incomes, assets and liabilities during the year commencing the date of listing. The companies were to undertake to, henceforth, pay their due taxes in full. Further, a tax concession of 5% was introduced for newly listed companies for 5 years post-listing, provided the firm lists a minimum of 30% of its fully issued and authorized share capital on the Nairobi Stock Exchange. This meant that newly-listed companies pay a corporate tax of 25% compared to 30% for unlisted firms.

_

¹² There was also an attempt to boost the market in Government bonds. The Government floated a Eurobond in 1998 in order to reduce domestic Government borrowing. In addition, all outstanding Government overdrafts with the Central Bank were converted into marketable securities.

4. Conclusion

This study traces the development of the Nairobi Stock Exchange since the initiation period in the 1920s, with a focus on institutional and policy changes. An evolutionary process with three stages of development is portrayed. The initiation stage is mainly characterized by no formal rules and regulations and is dominated by foreign investors. In the formalization stage, a self-regulatory system is adopted while attempts are made to increase the participation of local investors by the post-independent Government. In the third stage, various institutional and policy reforms are implemented to enhance the growth of the market.

Two main factors seem to shape the development path: the political environment both at the local level and in the East African region which saw change in the policy environment and changed the composition of market participation; and secondly the macroeconomic environment which instigated the demand for locally-mobilized long term capital to enhance economic development. The extent to which the development path has influenced the performance of the market is interesting to investigate.

Considering the developments in various aspects of the market, the following patterns are indicated. First, the stock brokerage industry has expanded and diversified with the number of stock brokers increasing from 6 to 20, and licensing of stock dealers and investment banks. However, stock brokerage is yet to be fully negotiable while the role of stock brokers remains predominately that of an agent. Second, the composition of market participants shows a shift from a market dominated by foreign investors in the initiation stage to increased participation of local investors in the formalization stage (especially in the post-independence period) and re-entry of foreign investors, though at limited level, and mass education on stock market operations and assets in the revitalization stage. Third, the NSE served as a regional

market for the East African states in the initial stages but the Stock Exchange lost a significant proportion of its market scope due to the political changes in these states. Attempts are however underway to establish a regional market to facilitate expansion of the NSE market. Fourth, though the diversity of securities traded is still minimal, the reform period saw efforts made to attain market depth with introduction of new instruments. This is a step forward in deepening the capital market. Fifth, while a discriminative tax policy that penalized share investors heavily was adopted in the post-independence period, an incentive-based tax policy regime was adopted in the reform period to enhance the competitiveness of the financial assets and reduce the barriers to listing new issues. Sixth, in an effort to strengthen the regulatory framework, the regulatory system has witnessed a shift from the non-formal system to self-regulatory and statutory regulatory framework. However, the effectiveness of the new regulatory framework will depend on the comprehensiveness of the regulatory framework and its enforceability. Seventh, the trading system shows a shift from the coffee-house forum to floor trading and there are attempts to introduce the delivery versus payment system with the introduction of the Central Depository System and automation of the trading cycle. The main aim is to enhance efficiency in the price discovery process and liquidity of the market. These changes mimic development paths of other stock markets in both the developed and emerging markets.

References

- Akamiokhor, G. (1996). The role of securities commissions in emerging stock capital markets. <u>In</u> Sam Mensah: *African capital markets: contemporary issues.* Rector Press, p142-160.
- Amihud, Y., H. Mendelson and B. Lauterbach (1997). "Market microstructure and securities values: evidence from Tel Aviv stock exchange". *Journal of Financial Economics*, 45, p365-390.
- Amihud, Y., Mendelson, H. amd M. Mwigia (1990). "Stock market microstructure and returns volatility: evidence from Italy". *Journal of Banking and Finance*, 14, p423-440.
- Aworolo, E.A. (1971). The development of capital markets in Africa with particular reference to Kenya and Nigeria. IMF Staff Papers, 118, p420-469.
- Bessembinder, H. and H. M. Kaufman (1997). "A cross exchange comparison of execution costs and information flow for NYSE listed stock". Journal of Financial Economics, 46(3), p293-320.
- CMA, Annual Reports (various issues).
- De Santis G. and Imrohoroglus (1997). "Stock returns and volatility in emerging financial markets". *Journal of International Money and Finance*, 16(4), p561-579.
- Garman, M.B. (1976). "Market microstructure". *Journal of Financial Economics*, 3, p257-275.
- Georgakopoulos, N.L.(1996). "Why should disclosure rules subsidize informed traders"? *International Review of Law and Economics*, 16, p417-431.
- Government of Kenya (1974). Development Plan 1974/78. Nairobi: Government Printer.
- Government of Kenya (1979). *Development Plan 1979/84*. Nairobi: Government Printer.
- Government of Kenya (1986). Sessional Paper No. 1 on economic management for renewed growth.

- Government of Kenya (1990). Capital Market Authority Act (Chapter 485A) revised version. Nairobi: Government Printer.
- Government of Kenya (various). *Economic Survey*. Nairobi: Government Printer.
- IFC/CBK (1984). Development of money and capital markets in Kenya. Nairobi: Government Printer.
- La Porta R., F. Lopez-de-Silanes and A. Shleifer (1998). "Law and Finance". *Journal of Political Economy*, 58, p1113-1155.
- La Porta R., F. Lopez-de-Silanes, A. Shleifer and R. Vishny (2000). "Investor protection and corporate governance". *Journal of Financial Economics*, 58, p3-27.
- Madhavan, A. (1992). "Trading mechanism in securities market". *Journal of Finance*, XLVII(2), p607-641.
- Munga, D.M.(1974). The Nairobi Stock Exchange: its history, organization and role in the Kenyan economy. Unpublished MBA Thesis, University of Nairobi.
- NSE (1997a). Annual report 1997. Nairobi: NSE.
- NSE (1997b). Rules and regulation manual. Nairobi: NSE.
- NSE (1997c). Listing manual. Nairobi: NSE.
- NSE (1998) Annual report 1998. Nairobi: NSE.
- Parkinson, J.N. (1984). "The Nairobi Stock Exchange in the context of development of Kenya". *Savings and Development*, VIII (4), p363-372.

Appendix: Chronology of events

Period Action taken

1964: Kampala agreement 1960s

> 1965: Dissolution of common currency area, passing of Central Bank Act, and exchange control on sterling transaction

1967: Banks in Tanzania are nationalized and EAC treaty is signed in December

Kenyanisation policy is implemented

1970s 1969/70: Lending to foreign-controlled company was raised as a concern and therefore

the exchange control. All companies pay their tax at the end of the accounting year to allow even flow of revenue and more prompt payment of tax

1970: Uganda imposes exchange controls on Kenya and Tanzania

1971: 12.5% tax on gross dividends and interest paid by corporation is imposed to control movement of funds out of Kenya. 20% tax on all fees and royalties payable to nonresidents

Capital Issue Committee is established 1973/74: stamp duty increases to 3%

1974/75: Dividend tax for non-residents increases from 12.5% to 15% and for residents from 12.5% to 20%.

June 1975: Capital gain tax of 36% imposed to generate Government revenue and curb excessive speculation

1975/76: Collapse of the EAC followed by de-listing of firms reducing the market scope

1980s IFC and CBK conduct a joint study on capital market development issues

June 1985 capital gain tax suspended

1990 CMA is established, constituted in January 1990 and inaugurated in March 1990

June 1991 Stamp duties on retail shares is abolished for listed firms

Transaction costs of public issues of shares, bonds and debentures made tax-deductible 15% withholding tax on dividend income made final and 10% for interest income, a iming to reduce double taxation on corporate dividends and incomes to individuals arising from investment in securities

NSE registered as a limited company

September 1991 CMA membership to IOSOC is granted as the 5th African country to secure

membership

July 1991

November 1991 Trading system shifted to floor trading

April 1992 Car and General reinstated to public trading list June 1992

Withholding tax on dividend income reduced to 10%, rationalizing the rate applicable

to interest income

December 1992 Trade Bank denied listing due to irregularities in disclosure management All restrictions on remittance of profits dividends and expatriates earning removed November 1993 February 1994 Capital controls for offshore borrowing relaxed subject to quantitative limits and

finalized in May 1994

Residents allowed to borrow up to maximum US\$ 1 million July 1994 Seven new stockbrokers licensed to make the market more competitive

Gazettement of CMA (Amendement) regulations 1994 and capital market (Amendment) rule 1994 as legal notice No. 232 and 233 in Kenya Gazette supplement No. 4 of 22 July 1994. These amendments require an exchange to be

a pproved by the CMA to be a company limited by guarantee

January 1995 Entry of foreign investors limited to 20%

The Capital Market Authority Amendment Act, 1994 is passed

The CMA (foreign investors) regulation 1995 are Gazetted in legal notice No. 3 in the Kenya Supplement No. 1 of 6 January, providing for foreign investors participation in the domestic capital market subject to a 2.5% limit for individual investors and 20% in aggregate for each stock

Bob Mathews stockbroker suspended from the exchange trading floor due to

irregularities in its operations

February 1995 Regional training course aimed at human capital development for technical staff of

ČMA

March 1995 Education trip to South East Asia of capital market staff

May 1995 CMA declines to renew the license of Bob Mathews stockbrokers for misconduct and gross violation of CMA Act and the CMA rules and regulations. CMA installs modern computer network to enhance its operations and facilitate information

collation, analysis and dissemination.

June 1995 Withholding tax reduces to 7.5% for local investors in quoted securities

Issue of commercial paper allowed for listed companies Government announces its support of the CDS and rating agency

Eight more stockbrokers are licensed

Board of directors of NSE is reconstituted

15 June: CMA (Amendments) regulations 1995 are issued allowing CMA to collect fees from the market to finance its operations in Legal Notice No. 212 of Kenya Gazette Supplement No. 39

Period	Action taken
July 1995	Compensation fund introduced to protect the investors against loses arising from equity trading
August 1995	Foreign investors participation is raised to 40% The CMA (Foreign Investor No. 2) regulations 1995 are issued in Legal Notice No. 291, Kenya Gazette Supplement
January 1996	No. 55 of 24 August 1995. CMA allowed levying charges on public issues, bonuses, rights and private placements by public companies and
1. 1.4007	gazetted in July 1997
March 1996 April 1996	Submitted a CDS proposal to the Attorney General for review Barring Asset Management set up a regional investment fund for Africa, Simba Fund, with a resource pool of US\$ 120
	million (Ksh 6.9 billion) invested in seven sub-Saharan African economies including Kenya
May 1996	Kenya National Oil Company Ltd (Kenol), following a resolution at the company AGM, applies to de-listed from the NSI
June 1996	As a result, trading in Kenol at the NSE is suspended. Withholding tax reduced to 5% for local investors and 7.5% for the non-residents and it remains final. This reducing do uble taxation on equity investment incomes of individuals was raised from 10% to 15% while withholding tax on
July 1996	bearer instruments and certificates of deposit was raised from 10% to 20% Establishment of the Central Depository System progresses with the drafting of the CDS Bill almost being finalized by
	a consultant Brian Taylor Associate of UK. Kenya Finance Bank is put under the statutory management of the Central Bank of Kenya after persistent liquidity crises and suspended from the NSE.
August 1996	CMA forms 11 member disclosure standard committee to review the existing disclosure requirements and make
	recommendations
	CMA approves amendments governing takeovers Establishment of compensation fund
	Uganda Stock Exchange has the first ever public floatation of shares in the privatization of Uganda Grain Milling
Cantambar 1006	Corporation where the Government offered for sale 49% to public and 51% to core investors
September 1996	CMA (Amendment) rules 1996 Gazetted providing additional requirements on takeovers and de-listings to protect minority shareholders, in Legal Notice No. 286 of Kenya Gazette supplement No. 52 of 13 September 1996
October 1996	Disclosure committee formed by CMA
	Central Bank liquidates Kenya Finance Bank and appoints the Depository Protection Fund liquidator The NSE records the biggest deal in its 42-year history with NIC bank selling 4.9 million shares valued at Ksh 217
	million The gradual privatization of Kenya Commercial Bank continues with the successful public offer for sale of 11.88
	million shares at Ksh 50 each representing 10% of the company's issued capital, which subsequently reduced
	Government shareholding to 60% and is targeted for 40% Kenya's first venture capital fund, the Acacia Fund Ltd is launched
	An offshore investment fund incorporated in the Cayman Islands, Regent Undervalued Assets Africa Fund applies for
November 1996	a simultaneous listing at the NSE, Gaborone (Botswana) and Dublin (Ireland) Kenol suspension is lifted
November 1996	The first corporate bond is listed (the EABD bond)
January 1997	A new CMA Board is inaugurated
February 1997	Government lists the first Treasury bond
March 1997	An association of stockbrokers is formed to develop code of conduct and promote professionalism of stockbrokers East Africa member state securities regulatory authorities is formed under the memorandum of understanding
	between the securities regulators of Kenya, Uganda and Tanzania
April 1997	Education program first seminar held
	Treasury bond trading commences The first issue of Government three one-year floating rate bonds
	TPS Serena issues 12 million new ordinary shares to the public at a price of Ksh 13 each
May 1997	The second public education seminar held
June 1997	CMA issues guidelines on issuance of corporate bonds, commercial paper and other debt instruments Tax exempt of gains by insurance companies from trading at the stock exchange aiming to release for trading large
•	stockholding which were in the custody of insurance companies and therefore boost the supply of securities in the market
August 1997	IMF freezes EASF aid facility to Kenya, weakening the shilling, reducing the foreign reserves, and increasing inflation
September 1997 October 1997	African Tours and Hotels limited shares are suspended from trading on the NSE The first issue of two year floating rate bond
	Kingfisher Company Ltd (formerly Phillips International Ltd) application for lifting of the suspension is declined and the company advised to meet all the listing requirements.
December 1997	The Association of Financial Analysts is launched
April 1998	The second multi-party elections are held in Kenya CMA publishes the guidelines on disclosure standard by listed companies to form audit committees of non-executive
	directors with knowledge of business and accountancy Government issues the second two year floating rate bond
	KCB floats the fourth privatization offer
June 1998	Tax amnesty bonds floated aimed at encouraging compliance with Income Tax Act
	In surance companies required to invest minimum 20% of their gross premium in Government securities Government plan to promulgate legislation to hasten establishment of collective investment schemes like mutual
	funds and unit trust

Period	Action taken
November 1998	New stock index AMMI 27 launched by Alico Asset Management
December 1998	ICDC offer rights issue raising Ksh 350 million
January 1999 January 2000	CMA issues guidelines to promote corporate governance Rights issue by Pan African Insurance
March 2000	On the 21st African Lakes opens trading to public becoming the first tech stock and first dual listed share
April 2000 May 2000	CDS Bill passes the second reading in Parliament 8 May: CMA launches the reform strategy for fundamental reorganization of Kenya's capital market
July 2000	Tuly: Reorganization program for the stock market is launched to cater for specific needs of different issuers Huly: Kenya, Uganda and Tanzania come a step closer to the regionalization of stock exchange with the signing of the report of the joint stock exchange task force on cross-border listing
	17 July: Bids from prospective suppliers of CDS and ATS are opened 28 July: Kenya gets readmitted to the community of bankable economies when IMF agreed to advance support for the budget
August 2000	1 August: new trading cycle T+5 is launched, that is a process of registration in an ambitious preparation exercise for the implementation of the CDS environment, together with the implementation of the reforms on the NSE Board
	11 August: Ministers for Finance from Kenya, Uganda and Tanzania propose the establishment of the Capital Markets Development Committee (CMDC) as a separate committee under EAC framework; this is endorsed during the 31 meeting by the Council of Ministers. This was aimed at harmonizing policies and proposals of importance to the capital markets from the three EAC countries
	18 August: NSE and CDSC announce the short list of CDS and ATS system vendors. CDS Act and CMA act a re passed by parliament and receive presidential assent paving way for the full implementation of the CDS.
September 2000	CFC financial services is licensed as the first dealer 25September: Final CDS bid proposals opened
October 2000	Long awaited institutional framework for the regulation of the retirement benefits sector is gazetted NSE becomes a member of the association of national numbering agencies, the global securities numbering
	a gency NSE and CDSC completes the selection of a vendor to supply ATS and CDS; that is efaSEMS and Consortia
	ASEA (African stock exchanges associations hold its first conference of the new millennium
December 2000 February 2001	20 December: Shelter Afrique launches a floating rate Ksh 350 M 5 February: The three NSE boards become operational
March 2001	Workshop to create awareness on debt instruments targeting the stockbrokers 27 March: EABL shares commence trading at the Uganda Stock exchange
May 2001	EADB launches a Ksh 2 billion medium term note
January 2001 June 2001	Review of the listing rules Safaricom bond launched Ksh 4 billion medium term note
	With the budget, all companies seeking listing on NSE to pay 27% corporate tax instead of 30% for the first three years of listing and also to have amnesty from previously undeclared profits on condition that full
October 2001	disclosure is maintained in the first one year after listing ICDC launches its public share issue
November 2001	14 November: MSC begins official trading
	20 November: SNG first right issue 30 November: EA community legislative assembly is launched by EA heads of state
December 2001 January 2002	20 November: ICDC floats additional new shares 2 January: CMA under section 11(3) (L) of CMA 200, cancelled listing of Theta group, Regent undervalued Asset Africa Fund, Pearl Dry cleaners, and Lonhro Motors. Also fixed securities for KPCU, Marshalls,
	Kenya Orchards, SNG, Chancery Investment, Lonrho Motors, hutching Biemer, and Pearl Dry cleaners
March 2002	Total launches its right issue Kenya Airways shares introduced on the official list of Ugand a Stock Exchange
April 2002	CMA announces the approval of new NSE trading and settlement rules with amendments where block trades
May 2002	were revised upwards from Ksh 3 million to 50-200 million 12 May: Consultative meeting in Nairobi hosted by NSE and CMA to discuss proposals of the committee for
	presentation to the finance ministers CFC Financial Services Limited allowed to operate as an Investment bank without taking deposits to provide
July 2002	investment advisory services Any new listed company with an initial listing of at least 30% of its share capital allowed enjoying a corporation tax rate of 25% for a period of five years
	New foreign investors' regulations where the locals have 25% minimum reserve of the issued share capital with the balance 75% becoming a free float for all classes of investors. This applies to IPOs and GOK privatizations
August 2002	Shareholder agreement for NSE, CMA, AKS, CMA Investor compensation fund and nine institutional investors through the Capital markets Challenge Fund came together as investors in Central Depository and Settlement Corporation (CDSC)
October 2002	Hak Securities allowed operating as an investment bank after fulfilling the requirements

Appendix: Summary of institutional and policy changes

	Initiation stage (1920s-1953)	Formalization stage	Revitalization stage (1989-1999)
	·	independence indigenization period policy	
Regulatory system	Informal regulatory system	Self-regulatory system	Statutory regulatory system
Trading system	Call auction coffee-ho	Call auction coffee-house forum trading system	Call auction trading floor forum trading system
Stockbrokers	Non-specialized	Specialized stockbrokers (6)	Specialized stockbrokers (20)
Foreign investors	Dominating foreign	Restricted foreign investors trading	Relaxation of the restrictions on
	Pinco con cracing		Torong His colors Harring
Control of listed	Dominating foreign	Listing of local controlled firms,	Locally controlled firms
firms	controlled listed	decreasing proportion of foreign	dominating
	firms	controlled listed firms	
Taxation policy	Non-controlled	Discriminative taxation policy	Incentive based taxation policy
	policy regime	regime	
Securities traded	Non-diversified		Introduction of bonds and
			commercial paper
Market scope	East African Community (EAC)	uity (EAC) No EAC	Proposals for EAC and beyond
		market	
Political regimes	Pre-independence	Post-independence	
Macroeconomic	Un-controlled regime	Controlled	Decontrolled Policy regime
policy regimes	•	policy regime	

KIPPRA PUBLICATIONS

Conference Proceedings

- Report of the proceedings of the AERC-KIPPRA World Trade Organization (WTO) Workshop, 2000
- Report of the proceedings of the International Conference on Finance and Development: Evidence and Policy Issues, 2001

Discussion Papers

- Njuguna S. Ndung'u (2000). The exchange rate and the interest rate differential in Kenya: a monetary and fiscal policy dilemma. KIPPRA DP No. 1
- Karingi, S. N. and Njuguna S. Ndung'u (2000). *Macro models of the Kenyan economy: a review*. KIPPRA DP No. 2
- Ronge, E. E. and H.O. Nyangito (2000). A review of Kenya's current industrialization policy. KIPPRA DP No. 3
- Nyangito, H.O. (2001). Delivery of services to smallholder coffee farmers and impacts on production under liberalization in Kenya. KIPPRA DP No. 4
- Njuguna S. Ndungu and R. W. Ngugi (2000). Banking sector interest rate spread in Kenya. KIPPRA DP No. 5
- Karingi, S.N., M.S. Kimenyi and Njuguna S. Ndung'u (2001). *Beer taxation in Kenya: an assessment*. KIPPRA DP No. 6
- Ikiara, M.M. (2001). Vision and long term development strategy for Kenya's tourism industry. KIPPRA DP No. 7
- Geda, A. and Njuguna S. Ndung'u (2001). Specifying and estimating partial equilibrium models for use in macro models: a road map for the KIPPRA-Treasury Macro Model. KIPPRA DP No. 8
- Geda, A., Niek de Jong, G. Mwabu and M.S. Kimenyi (2001). *Determinants of poverty in Kenya: household-level analysis*. KIPPRA DP No. 9
- Were, M., A. Geda, S.N. Karingi and Njuguna S. Ndungu (2001). *Kenya's exchange rate movement in a liberalized environment: an empirical analysis*. KIPPRA DP No. 10
- Huizinga, F., A. Geda, Njuguna S. Ndung'u and S.N. Karingi (2001). Theoretical base for the Kenya macro model: the KIPPRA-Treasury macro model. KIPPRA DP No. 11
- Mwabu, G., M. S. Kimenyi, P. Kimalu, N. Nafula and D. K. Manda (2002). Predicting household poverty: a methodological note with a Kenyan example. KIPPRA DP No. 12
- Manda, D.K., G. Mwabu, M. S. Kimenyi (2002). Human capital externalities and returns to education in Kenya. KIPPRA DP No. 13
- Bedi, A., P.K. Kimalu, D.K. Manda, N.N. Nafula (2002). The decline in primary school enrolment in Kenya. KIPPRA DP No. 14

- Odhiambo, W. and H. Nyangito (2002). Land laws and land use in Kenya: implications for agricultural development. DP No. 15
- Were, M. and S. Karingi (2002). Better understanding of the Kenyan economy: simulations from the KIPPRA-Treasury Macro Model. KIPPRA DP No. 16
- Nyangito, H., M. Ikiara and E. Ronge (2002). *Performance of Kenya's wheat industry and prospects for regional trade in wheat products*. DP No. 17
- Nyangito, H. and L. Ndirangu (2002). Impact of institutional and regulatory framework on the food crops subsector in Kenya: 1990-1999. KIPPRA DP No. 18
- Ikiara, M. (2002). Impact of tourism on environment in Kenya: status and policy. KIPPRA DP No. 19
- Ronge, E., L. Ndirangu and H. Nyangito (2002). Review of government policies for the promotion of micro and smallscale enterprises in Kenya. KIPPRA DP. No. 20
- Kiringai, J., Njuguna S. Ndung'u, and S.N. Karingi (2002). *Tobacco excise tax in Kenya: an appraisal*. KIPPRA DP No. 21
- Were, M., Njuguna S. Ndung'u, A. Geda and S.N. Karingi (2002). *Analysis of Kenya's export performance: an empirical evaluation*. KIPPRA DP No. 22
- Ikiara, M.M., L. Ndirangu (2003). Prospects of Kenya's clothing exports under AGOA after 2004. KIPPRA DP No. 24
- Nyangito, H. (2003). Agricultural trade reforms in Kenya under the WTO framework. KIPPRA DP No. 25
- Odhiambo, W. and H. Nyangito (2003). *Measuring agricultural productivity in Kenya: a review of approaches*. KIPPRA DP No. 26

Occassional Papers

- Gitu, K. W. (2001). Strengthening the link between policy research and implementation. KIPPRA OP No. 1
- Kimenyi, M.S. (2001). Effective private sector representation in policy formulation and implementation. KIPPRA OP No. 2
- Kimenyi, M.S. (2002). *Agriculture, economic growth and poverty reduction*. KIPPRA OP No. 3
- Nyangito, H. (2002). Post-Doha African challenges in the sanitary and phytosanitary and trade related intellectual property rights agreement. KIPPRA OP No. 4

Policy Papers

- Nyangito, H.O. (2001). Policy and legal framework for the tea subsector and the impact of liberalization in Kenya. KIPPRA PP No. 1
- Nyangito, H.O. (2001). Policy and legal framework for the coffee subsector and the impact of liberalization in Kenya. KIPPRA PP No. 2

Special Reports

- Legal and other constraints on access to financial services in Kenya: survey results. KIPPRA Private Sector Development Division. SR No. 1, 2001
- Thinking about regulating? The better regulation guide. KIPPRA Private Sector Development Division. SR No. 2, 2002
- Policy timeline and time series data for Kenya: an analytical data compendium. KIPPRA Macroeconomics Division, SR No. 3, 2002

Working Papers

- Wasike, W.S.K. (2001). Road infrastructure policies in Kenya: historical trends and current challenges. KIPPRA WP No. 1
- Ikiara, M.M. (2001). Policy framework of Kenya's tourism sector since independence and emerging policy concerns. KIPPRA WP No. 2
- Manda, D.K., M.S. Kimenyi and G. Mwabu. *A review of poverty and antipoverty initiatives in Kenya*. KIPPRA WP No. 3
- Kimalu, P.K., N. Nafula, D.K. Manda, G. Mwabu and M.S. Kimenyi (2001). *Education indicators in Kenya*. KIPPRA WP No. 4
- Geda, A., S.N. Karingi, Njuguna S. Ndung'u, M. van Schaaijk, M. Were, W. Wassala and J. Obere (2001). Estimation procedure and estimated results of the KIPPRA-Treasury macro model. KIPPRA WP No. 5
- Kimalu, P., N. Nafula, D.K. Manda, G. Mwabu and M.S. Kimenyi (2002). *A situational analysis of poverty in Kenya*. KIPPRA WP No. 6
- Kiringai, J. and G. West (2002). Budget reforms and the Medium-Term Expenditure Framework in Kenya. KIPPRA WP No. 7
- Ikiara, M. and L. Ndirangu (2003). Developing a revival strategy for Kenya's cottontextile industry: a value chain approach. KIPPRA WP No. 8
- Ng'eno, N.K., H.O. Nyangito, M.M. Ikiara, E.E. Ronge, J. Nyamunga (2003). Regional integration study of East Africa: the case of Kenya. KIPPRA WP No. 9
- Manda, D. K., P.K. Kimalu, N. Nafula, Diana K. Kimani, R. K. Nyaga, J.M. Mutua, G. Mwabu, M.S. Kimenyi (2003). *Cost and benefits of eliminating child labour in Kenya*. KIPPRA Working Paper No. 10